Author's response to reviews

Title: Gastroprotective effect of Desmosdumotin C isolated from Mitrella Kentii against ethanol-induced gastric mucosal hemorrhage in rats: possible involvement of glutathione, heat-shock protein-70, sulphhydryl compounds, nitric oxide and anti-helicobacter pylori activity

Authors:

Heyam M A Sidahmed (diamondhm@hotmail.com)  
Ainnul Azizan (ainnol_azizan@yahoo.com)  
Syam Mohan (syammohan@yahoo.com)  
Mahmood A Abdulla (mahmood955@yahoo.com)  
Siddig I Abdelwahab (Siddigroa@yahoo.com)  
Manal M E Taha (manalroa@yahoo.com)  
A. Hamid A. Hadi (ahamid@um.edu.my.com)  
Kamal A Ketuly (KKetuly@yahoo.com)  
Najihah M Hashim (najihahmh@um.edu.my)  
Mun F Loke (lokemunfai@um.edu.my)  
Jamuna Vadivelu (jamuna@um.edu.my)

Version: 3 Date: 30 November 2012

Author's response to reviews: see over
Subject: Submission of manuscript

Dear Editor,

I would like to submit the following manuscript titled “Gastroprotective effect of Desmosdumotin C isolated from Mitrella Kentii against ethanol-induced gastric mucosal hemorrhage in rats: possible involvement of glutathione, heat-shock protein-70, sulphhydryl compounds, nitric oxide, anti-cyclooxygenase-2 and anti-helicobacter pylori activity” for possible evaluation. Our findings are unique and giving sufficient understanding on how this compound works on ulcer. I affirm that the manuscript has been prepared in accordance with Instructions to Authors. I have read the manuscript and I hereby confirm that the content of this manuscript thereof has not been published in a refereed journal, and it is not being submitted for publication elsewhere. I declare that all the listed authors have read and approved the submitted manuscript and that the present study was performed according to international, national and institutional rules considering animal experiments, clinical studies and biodiversity rights. In the revised version manuscript we clarify that all procedures relating to animal care and the animal research protocols conformed to the animal care guidelines of the Institutional Animal Care and Use Committee, University of Malaya. This study specifically was presented to the institutional ethical review board (UM ICUCA) for approval, and the approval was granted [Ethic No FAR/29/06/2012/HMAS(R)].

Kind regards

Dr Heyam Mohamed Ali Sidahmed
Corresponding Author